Author's response to reviews

Title: Computerised Therapy for Depression with Clinician vs. Assistant and Brief vs. Extended Phone Support: Study Protocol for a Randomised Controlled Trial

Authors:

Lina Gega Dr (l.gega@uea.ac.uk)
Louise Swift Dr (l.swift@uea.ac.uk)
Gary Barton Dr (g.barton@uea.ac.uk)
Gillian Todd Dr (g.todd@uea.ac.uk)
Nesta Reeve Ms (nesta.reeve@nsft.nhs.uk)
Kelly Bird Ms (kelly.bird@nsft.nhs.uk)
Richard Holland Prof (r.holland@uea.ac.uk)
Amanda Howe Prof (A.howe@uea.ac.uk)
Jon Wilson Dr (jonathan.wilson@nsft.nhs.uk)
Jo Molle Ms (J.Molle@uea.ac.uk)

Version: 3  Date: 22 July 2012

Author's response to reviews:

We have slightly amended the title to conform to journal style for study protocol articles and we have made some small formatting changes as per author instructions.